Author’s response to reviews

Title: Medication management support in diabetes: A systematic assessment of diabetes self-management apps

Authors:

Zhilian Huang (ZHUANG014@e.ntu.edu.sg)
Elaine Lum (elaine_lum@ntu.edu.sg)
Geronimo Jimenez (geronimo.jimenez@ntu.edu.sg)
Monika Semwal (monikasharma11@gmail.com)
Peter Sloot (psloot@ntu.edu.sg)
Josip Car (josip.car@ntu.edu.sg)

Version: 2 Date: 06 Jun 2019

Author’s response to reviews:

Dear Dr Lee,

Revision and resubmission of manuscript ID: BMED-D-19-00479 (Medication management support in diabetes: A systematic assessment of diabetes self-management apps)

Thank you for your decision to accept our publication.

We have addressed the editorial comments and detailed our responses together with manuscript changes in the enclosed table (uploaded to the system). The references mentioned under the “manuscript changes” section have been included in our revised manuscript.

Thank you.

Best wishes,

(Ms) Zhilian Huang

On behalf of all authors